VC/Week 37 starting 11sept 06
SM18 Operation Meeting (Tuesdays) 12 sept 06

Bench status  on Monday  11-09-2006 at 9/11/2006 9:08:58 AM
	8
	A1
	MBAR2274
	Prep 5 Cool Down
	99.93
	Mistake not have done ROBTC 
	
	Last time

	4
	A2
	MBAR1338
	PT 12 Warm Up
	13.25
	
	
	

	5
	B1
	SSS329
	PT 6.1 Training 1
	1.93
	
	
	

	1
	B2
	SSS309
	PT 13.2 Resist. Meas.
	298.58
	
	
	

	7
	C1
	SSS298
	Prep2.2C (CD clamp)
	297.41
	HV fail- not trained
	
	

	3
	C2
	SSS339
	PT 6.6 Training 6
	4.51 HV
	
	
	

	2
	D1
	SSS526
	PT 3 QH Measure
	1.91
	ROBTC
	
	

	10
	D2
	SSS608
	Prep 3 CDW1 HV
	298.76
	YT318 no antC
	
	

	9
	E1
	MBAR1374
	Prep 5 Cool Down
	262.41
	
	
	

	11
	E2
	MBAL1401
	ICS 3 Traveller after warm test
	299.2
	
	
	

	6
	F1
	MBAL2168
	PT 13.2 Resist. Meas.
	298.58
	For cryo leak
	
	

	12
	F2
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	298.58
	
	
	


 
  
CONFIG :  D 6 - AS 3 -  IRSSS 3 ;   
Supervisor:  vittorio  ;Eq support : federico 16-3602 
17 Magnet Tests completed (  ) 
9 Dipoles tests  ( 1369 ROBTC)


6 Arc-SSS  tests ( 309 put tro ROBTC .. rest stripped)
 1  500-series [ 501 (193 hr) ]
1  600-series [  Q6 680 (214 h) , 6 MQTLs  ]
General Points:
· News /Situation  : D1 608 retest  for YT318. No Anticryo. Faster Training now ?
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· Situation SSSS  :D1 & D2 issues; 
· A.O.B..
· Problems last Thurs/fri on B IAP at warm /disconnect/check/  : Federico
· D1 D2 switching issues wed & problems (Giorgio/salem) 
· + later on Federeico corr powering on D
	Magnets under Test from 03-09-2006 to 09-09-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR2379 
	[image: image3.png]



	


	02-09-2006 
	35 h.
	41 h.
	25 h.
	18 h.
	32 h.
	09-09-2006 
	151 h.
	Stripping
	
	

	2
	TBA2
	MBBR1369 
	[image: image5.png]



	


	31-08-2006 
	28 h.
	19 h.
	35 h.
	18 h.
	0 h.
	04-09-2006 
	100 h.
	Not Found or Thermal cycle
	
	

	3
	TBA2
	MBBR1369 
	[image: image7.png]



	


	04-09-2006 
	1 h.
	19 h.
	12 h.
	29 h.
	11 h.
	07-09-2006 
	72 h.
	Stripping
	
	

	4
	TBB1
	SSS305 
	[image: image9.png]



	


	31-08-2006 
	25 h.
	11 h.
	35 h.
	8 h.
	23 h.
	04-09-2006 
	102 h.
	Stripping
	
	

	5
	TBB2
	SSS309 
	[image: image11.png]



	


	05-09-2006 
	22 h.
	12 h.
	47 h.
	5 h.
	0 h.
	08-09-2006 
	86 h.
	Not Found or Thermal cycle
	
	

	6
	TBB2
	SSS310 
	[image: image13.png]



	


	30-08-2006 
	49 h.
	13 h.
	26 h.
	7 h.
	39 h.
	04-09-2006 
	134 h.
	Stripping
	
	

	7
	TBC1
	SSS315 
	[image: image15.png]



	


	01-09-2006 
	79 h.
	24 h.
	53 h.
	8 h.
	24 h.
	09-09-2006 
	188 h.
	Stripping
	
	

	8
	TBC2
	SSS241 
	[image: image17.png]



	


	28-08-2006 
	34 h.
	15 h.
	16 h.
	6 h.
	77 h.
	03-09-2006 
	148 h.
	Stripping
	
	

	9
	TBC2
	SSS346 
	[image: image19.png]



	


	03-09-2006 
	22 h.
	29 h.
	33 h.
	10 h.
	9 h.
	08-09-2006 
	103 h.
	Stripping
	
	

	10
	TBD1
	SSS501 
	[image: image21.png]



	


	27-08-2006 
	87 h.
	17 h.
	28 h.
	9 h.
	52 h.
	04-09-2006 
	193 h.
	Stand By
	
	

	11
	TBD2
	SSS680 
	[image: image23.png]



	


	30-08-2006 
	35 h.
	15 h.
	101 h.
	13 h.
	50 h.
	08-09-2006 
	214 h.
	Stand By
	
	

	12
	TBE1
	MBAL1386 
	[image: image25.png]



	


	04-09-2006 
	36 h.
	27 h.
	18 h.
	15 h.
	27 h.
	09-09-2006 
	123 h.
	Stripping
	
	

	13
	TBE1
	MBAL1399 
	[image: image27.png]



	


	30-08-2006 
	30 h.
	23 h.
	16 h.
	15 h.
	17 h.
	04-09-2006 
	101 h.
	Stripping
	
	

	14
	TBE2
	MBAR2376 
	[image: image29.png]



	


	31-08-2006 
	45 h.
	21 h.
	13 h.
	14 h.
	27 h.
	05-09-2006 
	120 h.
	Stand By
	
	

	15
	TBF1
	MBAL1383 
	[image: image31.png]



	


	29-08-2006 
	31 h.
	24 h.
	19 h.
	14 h.
	34 h.
	03-09-2006 
	122 h.
	Stripping
	
	

	16
	TBF1
	MBBR2360 
	[image: image33.png]



	


	03-09-2006 
	24 h.
	22 h.
	24 h.
	16 h.
	8 h.
	07-09-2006 
	94 h.
	Stripping
	
	

	17
	TBF2
	MBBR1343 
	[image: image35.png]



	


	01-09-2006 
	27 h.
	21 h.
	14 h.
	15 h.
	11 h.
	05-09-2006 
	88 h.
	Stripping
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR2379
	11369 A.
(D1_L)
	11862.5 A.
(D1_L)
	12600 A.
(D2_L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBR1369
	11605.7 A.
(D2_U)
	12272.1 A.
(D2_L)
	11998 A.
(D1_L)
	12241.5 A.
(D1_L)
	12449.5 A.
(D2_U)
	12360 A.
(D2-U)
	12641 A.
(D2-U)
	12634 A.
(D2-U)
	12722 A.
(D2_U)
	[image: image38.png]



	-
	-
	-

	3
	MBBR1369
	12470.4 A.
(D1_L)
	12723 A.
(D2-U)
	[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS305
	12120.9 A.
(QI_12)
	12238 A.
(QE_34)
	12311.2 A.
(QE_34)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS309
	12430 A.
(QE-34)
	12591 A.
(QE_12)
	12824.5 A.
(QE_12)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS310
	9837 A.
(VClamp E)
	12735.9 A.
(QE_34)
	12801 A.
(QE-12)
	11241 A.
(VClamp I)
	11241 A.
(VClamp I)
	12816 A.
(QE_12)
	[image: image42.png]



	-
	-
	-
	-
	-
	-

	7
	SSS315
	9225 A.
(QE-34)
	12137.8 A.
(QI_12)
	12313 A.
(QI_34)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS241
	11169.3 A.
(QE_34)
	12207.2 A.
(QE_12)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS346
	11770.9 A.
(QI_34)
	11938.5 A.
(QE_34)
	12179.7 A.
(QE_34)
	12350.6 A.
(QE_34)
	12577.5 A.
(QE_12)
	12715.8 A.
(QI_34)
	12827 A.
(QE_12)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-

	10
	SSS501
	12300.1 A.
(QE_12)
	12424.1 A.
(QE_12)
	12656.7 A.
(QI_12)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS680
	316 A.
(No Data)
	395 A.
(No Data)
	433 A.
(No Data)
	428 A.
(No Data)
	443 A.
(No Data)
	467 A.
(No Data)
	347 A.
(No Data)
	472 A.
(No Data)
	478 A.
(No Data)
	474 A.
(No Data)
	477 A.
(No Data)
	483 A.
(No Data)
	384 A.
(No Data)
	412 A.
(No Data)
	369 A.
(No Data)
	484 A.
(No Data)
	475 A.
(No Data)
	0 A.
(No Data)
	248 A.
(No Data)
	296 A.
(No Data)
	475 A.
(No Data)
	490 A.
(No Data)
	473 A.
(No Data)
	[image: image47.png]




	12
	MBAL1386
	11342.4 A.
(D2_U)
	12395.8 A.
(D1_L)
	12700.8 A.
(D2-L)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	
	
	

	13
	MBAL1399
	11959.1 A.
(D2_L)
	12556.5 A.
(D1_L)
	12704.3 A.
(D2_L)
	[image: image49.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	
	
	

	14
	MBAR2376
	11883 A.
(D1-U)
	12377.8 A.
(D1_L)
	12850 A.
[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	
	
	

	15
	MBAL1383
	11083.3 A.
(D2-U)
	12466.7 A.
(D1-L)
	12779.9 A.
(D1_U)
	[image: image51.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	
	
	

	16
	MBBR2360
	11331.7 A.
(D1_U)
	12466.7 A.
(D2_U)
	12850 A.
[image: image52.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	
	
	

	17
	MBBR1343
	10917.1 A.
(D1_L)
	12117 A.
(D2_L)
	12758.8 A.
(D2_U)
	[image: image53.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
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Long Training + extra tests due to anticryostat on 680 with 6 MQTL’s   .. also additional questionable tests with antic sealed

	Date
	Bench
	Problem
	Type
	Application/
soft
	Action/advise

	Tuesday_05_A
	TBD2
	SSS680: Message received "Anticryostat Temperature too low". AC cable not connected. 
	MTM
	
	Guy informed. He suggested proceeding ignoring the message.

	Tuesday_05_N
	TBD1
	SSS526: Clamp correctors check could not be done Permanent triggers due to the existing settings done by Giorgio to make the electronics compatible to the magnet s4 - 680. s.
	MTM
	IAP
	Contacted Guy. He is not aware of the changes made by Giorgio. Awaiting intervention of Giorgio/mathias

	Tuesday_05_N
	TBC2
	SSS-346: Corrector powering on IJ could not be carried out as there is a note on the Power converter (IJ) instructing not to use the power supply. 
	AB/PO
	
	IJ training will be carried out after the converted is rectified and carrying on with training. In fact this PC is no longer used for corrector test but the adjacent one.
Richard comments: I have put some stickers on the rack with the spare 60Amps power converter. The ones we are using, to power cluster C, are the ones with the switch on top of them. It has always been like that.

	Tuesday_05_N
	TBB1
	SSS329 : Labels missing (for 1 and 2) on voltage tap connection leads. 
	CRI
	
	ICS had problem to perform the connection.

	Tuesday_05_N
	TBC2
	Multimeter is not displaying the last digit.
	
	
	The multimeter was set in fast sampling instead of medium sampling mode

	Wednesday_06_A
	TBD2
	SSS680 : Final HV is done and sign out E-traveller. Paper traveller pages not available.
	MTM
	
	

	Wednesday_06_N
	TBD1
	SSS5326: IAP test in fault Vano and V cathode.
	MTM
	
	Federico informed. He asked to proceed.

	Wednesday_06_N
	TBE1
	MB1386: HV test at cold BP failed: dipole_quench-heaters vs ground.
	MTM
	
	Federico came and check that all the circuits are OK separately. He advised to perform PT3, PT4 and PT5 and then check again.

	Wednesday_06_N
	TBB2

TBC2
	SSS309 and SSS346: TCL_aux not OK


	ACR
	
	ACR team informed. Team leater advised to change the magnet priority to get the condition in TBB2.

	Thursday_07_M
	TBA2
	MB1369: Magnet held up at 290 K for7 hours.

	MTM
	
	4 magnets warming up and MB1369 in low priority. Priorities changed.

	Thursday_07_M
	TBB2
	SSS309: Again training quench in the clamp. Still problem of flow in current leads

	ACR
	
	Cryo team takse the control in manuel mode.

	Thursday_07_M
	TBA2
	MB1369: IAP@warm. Offset error on VT_Cathode signal (Channel labelled VT_Anode on the potaim card).

	MTM
	
	Federico asked to try with another card. OK. Federico changed the faulty card.

	Thursday_07_A
	TBC1

TBB2
	SSS315: It takes a long time too cool down to 1.9 K
SSS309: Difficulty to stabilize He flow in current leads
	ACR
	
	

	Thursday_07_A
	TBB2
	SSS309: During PT8 quench is occurred in V-Clamp Ext near 12 KA current.
	ACR
	
	Federico informed to re-launch the test with increased Helium flow and during current ramping observe the V-Clamp voltage.
Cryo team asked to OP team to check if condensation is on the top of CFB. OP team put a blower.

The MQ quenched again in V_clamp after 3 mins.

Cryo stability reached in the night. PT8 performed.

	Thursday_07_N
	TBB2
	SSS309: HV CDAP failed due to the condensation in the CFB (high level flow) despite the blower.
All steps were ok except Quadrapole Vs M(X)D-M(X)F correctors & M(X)D-M(X)F correctors Vs Gnd failed.
	ACR
	
	PT9 executed manually.
Federico asked to open a NC (failure at 1.5 kV) 

	Thursday_07_N
	TBC1
	SSS315: HV (PT9) can not be launched due to LOT of Condensation on CFB. Humidity count = 68%..
	ACR?

MTM?


	
	Cryo team informed. All the parameters are OK from their side

	Friday_08_M
	TBB1
	SSS329: Cryo instability around 1.9 K
	ACR
	
	Cryo tema tried to solve the problem

Action ACR : pump at warm in the “decanteur” to remove eventual humidity.

	Friday_08_A
	TBB1

TBD1
	SSS329, SSS526 : Triggers during IAP at cold “ Vclamp_ext and V_supra”


	MTM
	
	Federico brigged the Vtaps faulty.

	Friday_08_A
	TBF2
	MB1400 : TRU1 had a pin missing in the connector (ap2).
	MTM
	
	ICS changed TRU1 with TRU5.

	Friday_08_A

Saturday_09_A

Saturday_09_A
	TBB1
	SSS329: Quench during the 1kA ramp due to VClamp.( Vtaps broken)
	ICS
	
	Called Federico, TC and A.Siemko for advice. A.Siemko advised to warm_up the magnet and ICS will  repeat the connection. Warm tests has to be performed before re-opening. Federico will inspect the V_taps during the opening.
Manchette re-open: A.Siemko and Federico  found an  inversion between two V_taps and the wires were not soldered carefully. ICS has redone the connection.

	Friday_08_N


	TBD1
	SSS526: Corrector cannot be performed.

During powering of CD/EF circuits the current ramp down at 13 A
	MTM
	
	Federico advised to train before the MQ. He will take action tomorrow.

Federico and Siemko cleared a faulty status on the energy extraction system of the correctors.
Federico called Sunday_morning. Extraction switch re-open

Monday Giorgio closed the switch and un validated the command



	Saturday_09_M
	TBD2
	SSS608: HWREC is off. TEMA read the previous magnet name SSS680.


	AB/CO


	HWREC
	Federico will take action.



	Saturday_09_A
	TBB2
	SSS309: HV CQBP OK after thermal cycle.
	MTM
	
	A.Siemko contacted for further instructions. He advised to continue the training with 2 quenches and warm_up with a training quench.

	Saturday_09_A
	TBF2
	MB1400: Warnings during MM in the angle differences (between backward and forward rotation)
	MTM
	
	Federico could not touch Stephane by phone. He advised to proceed.

	Sunday_10_M
	TBA1
	MB2274: Alarm message from ACR “temperature of anticryostats too low”
	
	
	

	Sunday_10_M
	TBA2
	MB1338: Quench at 1.5 kA for hf heaters: value of time discharge smaller in the graph than observed during the quench.

Same results but attenuated.
	AB/CO
	AQA
	Federico suggested trying again with manuel trigger.of QH card 2.

Federico suggested informing TC for decision.

	Sunday_10_A
	TBB
	The PC converter switch did not work : no light glowing
	AB/PO
	
	Federico asked to do it manually. OP TEAM managed to put the switch in CC after many trials

	Sunday_10_A
	TBC2
	SSS339: AQA analysis for 1st training could not be done as there was a buffer problem in DAQ server
	
	
	Contacted Federico. He suggested generating file by manually triggering after DAQ arming.

	Monday_11_M
	TBD1
	SSS526: IAP at cold (PT-1) was failing and there were offset error on QE_34-12,Q_I_E_lg, Vgas_Ext_lead
	MTM
	IAP
	Offsets measured in wrong outputs. Federico solved the problem.

































































